Author’s response to reviews

Title: A phase I/II study of weekly nab-paclitaxel plus cisplatin in chemotherapy-naïve patients with advanced non-small-cell lung cancer

Authors:

Yoshihiro Hattori (hattori@hp.pref.hyogo.jp)
Yuko Kono (kono119@hp.pref.hyogo.jp)
Shoichi Itoh (sho1.ito@hp.pref.hyogo.jp)
Takako Inoue (okuyama-ta@mc.pref.osaka.jp)
Yoshiko Urata (urata@hp.pref.hyogo.jp)
Yoshitaka Kawa (ohayoubird@gmail.com)
Rie Tohnai (tohnair@hyogo-cc.jp)
Toru Kumagai (kumagai-to@mc.pref.osaka.jp)
Kazumi Nishino (nisino-ka@mc.pref.osaka.jp)
Ryuji Uozumi (uozumi@kuhp.kyoto-u.ac.jp)
Satoshi Morita (smorita@kuhp.kyoto-u.ac.jp)
Shunichi Negoro (negoro@takarazuka-cst.jp)
Fumio Imamura (imamura-fu@mc.pref.osaka.jp)
Miyako Satouchi (satouchi@hp.pref.hyogo.jp)

Version: 2 Date: 22 Jan 2020

Author’s response to reviews:

Editor Comments:

1. We note that part of your manuscript has been previously presented as an abstract here: https://library.iaslc.org/search?search_location=Hall+1
While we are happy for manuscripts to have been partly presented as conference abstracts or posters, we do require this to be cited and acknowledged in the manuscript. Please therefore mention this previous publication in the 'Acknowledgements' section.

We added comments regarding these points.

(Acknowledgements section, line 10-13, page 12)

2. Please include the full names of the ethics committees of each participating centre (and the institute to which it belongs) that approved the study and the committee’s reference number if appropriate.

We added comments regarding these points.

(Ethics approval and consent to participate section, line 26-28, page 10)

3. Please ensure all figures are cited in the main manuscript text - currently Figure 3 is missing.

We revised these points.

(Methods section, line 5, page 5)

(Results section, line 27-29 and 31, page 7)

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

We uploaded our manuscript as clean version.
Reviewers’ comments:

Giorgio Scagliotti (Reviewer 1): The authors have adequately addressed the major critical issues, thus the paper may be now suitable for publication. I only suggest a final English language revision.

Thank you for the peer review and for the suggestion. English language revision was performed by American Journal Experts.

Vittorio Gebbia (Reviewer 2): Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated versions of manuscripts, graphs, supporting materials or other aspects of your report which cannot be included in a text format.

Please overwrite this text when adding your comments to the authors.

The paper is now acceptable

Thank you for the peer review.